Author’s response to reviews

Title: What impact do posters have on academic knowledge transfer? A pilot survey on author attitudes and experiences

Authors:

Nicholas Rowe (nrowe@bournemouth.ac.uk)
Dragan Ilic (dragan.ilic@med.monash.edu.au)

Version: 6 Date: 29 November 2009

Author’s response to reviews:

Dear Editors,

Please find attached our revised manuscript. We have amended the manuscript according to your requests;

Major Revisions
1. We have inserted a statement indicating that all authors have read and approved the final manuscript under 'Author’s Contributions'
2. We have modified Table 1 and 2 according to the instructions.

Minor Revisions
1. We have inserted the city in the affiliations section.
2. We have renamed 'Introduction' to 'Background'.
3. We have removed the funding statement to the 'Acknowledgements' section.
4. We have provided full details for references #6 and #16.

Many thanks,
Dragan Ilic